Author’s response to reviews

Title: Social determinants of overweight and obesity in the mother-child binomial: Evidences from Mexico

Authors:
Armando Arredondo (armando.arredondo@insp.mx)
CHRISTIAN PAUL TORRES DE LA ROSA (ciss09@insp.mx)
Emanuel Orozco (emanuel.orozco@insp.mx)
Oscar Resendiz (benjamin.nut.osc@gmail.com)

Version: 1 Date: 14 Apr 2020

Author’s response to reviews:

Dear Editor Archives of Public Health

By means of this letter I would like to highlight the changes made to the manuscript entitled "Social determinants of overweight and obesity in the mother-child binomial: Evidences from Mexico".

1. The authors have not reported the variables used and their controls within the logistic regression model.
We made the adjustment in the methodology section, explaining the control variables such as maternal BMI and the adjustment variables age, educational level, family income, food security, social security, diet patterns, smoking and alcoholism.

2. P values were reported inconsistently throughout the paper and should be revised, along with the inclusion of confidence intervals.
We made the correction of the P values in the results section, discussion and the tables that we made, in addition the 95% confidence intervals were added.

3. In order to increase the level of interest this article will receive from international audiences, I would recommend the authors review the paper for spelling/grammar and readability.

4. It may be useful to report income in both dlls and USD to provide context for these readers. The economic income from the head of the family, from the mother, from social programs, remittances, became dollars for a better understanding.

5. Additionally, the authors should review the referencing style used as an appropriate formatting style has not been applied.
We correct the bibliographic references according to the guidelines of the magazine. We consulted the revision of the references and corrected the invalid ones.

6. Please follow one of the statement of the Equator network for the redaction of the manuscript. Please revise the English (in the text and in the table).

This version of the paper was completely revised and edited by a professional translator who is a native English speaker and an expert in health related topics.

Sincerely yours,

MPH. Christian Paul Torres
Corresponding author